Author’s response to reviews

Title: Life-Threatening Bleeding from Gastric Dieulafoy's Lesion in A Pregnant Woman with HELLP Syndrome: A Case Report and Literature Review

Authors:

chen si (css81729@163.com)
Zhu Xiuli (1216243099@qq.com)
Jia Yong (727233947@qq.com)
Zhou Ying (13865901025@139.com)
Xie Li (13083006299@163.com)
Zhang Kaiguang (zhangkaiguang0097@163.com)

Version: 1 Date: 19 Mar 2017

Author’s response to reviews:

Dear Editor:

I have presented a detailed response to each reviewer point raised, made the necessary corrections and submit a revised manuscript online. There is no any changes to the authors, or order of authors, in my manuscript.

(Reviewer 1):

Dear Reviewer Hung:

Thanks for your sincere comments on my submitted manuscript.
1. According to your opinion, I tried to use more concise language and revised the entire manuscript with the help of an American doctor. So did I correct some grammatical errors.

2. I have presented the abbreviation of Dieulafoy's lesion (DL) after Dieulafoy's lesion.

3. The term "upper gastroscopy" was been replaced with esophagastroduodenoscopy (EGD) in case report (line 42-43).

4. I have removed the first horizontal lines in table 1.

5. I have presented the IRB approval letter from our hospital ethical committee which was saved as attachments.

(Reviewer 2):

Dear Reviewer Kao:

Thank you for sincere comments on my submitted manuscript.

I agree with you that diagnosis of HELLP syndrome may be influenced by other factors. In some condition, such as hypovolemic shock, hepatic changes with increased levels of enzymes may occur. Peripheral blood smear elevated, indirect bilirubin (38.5 μmol/L) and hypertension (155/91 mmHg) was found in current case. Laboratory results were consistent with the diagnosis of HELLP syndrome. So the pregnant woman was diagnosed with HELLP syndrome.